Author's response to reviews

**Title:** Delivering an Optimised Behavioural Intervention (OBI) to people with low back pain with high psychological risk; results and lessons learnt from a feasibility randomised controlled trial of Contextual Cognitive Behavioural Therapy (CCBT) vs. Physiotherapy

**Authors:**

- Tamar Pincus (t.pincus@rhul.ac.uk)
- Shamaila Anwar (s.t.anwar@leeds.ac.uk)
- Lance McCracken (lance.mccracken@kcl.ac.uk)
- Alison McGregor (a.mcgregor@imperial.ac.uk)
- Liz Graham (e.h.graham@leeds.ac.uk)
- Michelle Collins (m.collinson@leeds.ac.uk)
- John McBeth (j.mcbeth@keele.ac.uk)
- Paul Watson (pjw25@le.ac.uk)
- Stephen Morley (s.j.morley@leeds.ac.uk)
- Juliet Henderson (juliet.henderson@rhul.ac.uk)
- Amanda Farrin (juliet.hena.j.farrin@leeds.ac.ukc.uk)

**Version:** 6  **Date:** 18 May 2015

**Author's response to reviews:** see over
To the editor,

We have now added the TRN date of the trial to the TRN at the end of the abstract. We have inserted figure legend at the end of the manuscript, and have removed tables 1.1-1.4 from the figure. These are no presented as additional files.

Best wishes

On behalf of the OBI team

Tamar Pincus

Professor in Health Psychology